Author's response to reviews

Title: The Centella asiatica Juice effects on DNA damage, Apoptosis and Gene Expression in Hepatocellular carcinoma (HCC)

Authors:

Faridah Hussin (dka_as@yahoo.com)
Sima Ataollahi Eshkoor (simaataolah@yahoo.com)
Asmah Rahmat (satirann@yahoo.com)
Fauziah Othman (s1_ataolah@yahoo.com)
Abdah Akim (s1_ataolah@yahoo.com)

Version: 9 Date: 18 November 2013

Author's response to reviews:

Dear Reviewers,

I am so thankful for giving valued suggestions, which improve our paper. I already checked your comments and tried to find suitable answers. Hope they are adjusted to your questions. Meanwhile, English writing was checked as well. However, if there is anything else needed to be added please let me know.

Comments
1. Collecting place of plant and deposition place was mentioned
   Deposited number: Unfortunately not available
2. Figure legends were removed from figures
3. Authors contribution was checked and was adjusted

Sincerely Yours
Prof. Asmah Rahmat